**Author’s response to reviews**

**Title:** Knowledge of tuberculosis among female sex workers in Rajshahi city, Bangladesh: a cross sectional study

**Authors:**

Md. Masud Rana (masud.raj82@gmail.com)
Md. Rafiqul Islam (rafique_pops@yahoo.com)
Sheikh Moin Uddin (moin.lipu@gmail.com)
Md. Abdul Wadood (nazibw@gmail.com)
Md. Golam Hossain (hossain95@yahoo.com)

**Version:** 2 **Date:** 30 Sep 2019

**Author’s response to reviews:**

Response to Editor and Reviewers

Journal Name: BMC Infectious Diseases

Tracking No. (Manuscript ID): INFD-D-19-01271R

Manuscript Title: Old: "Survey on knowledge of tuberculosis among female sex workers in Rajshahi city, Bangladesh: a cross sectional study"

New Title: "Knowledge of tuberculosis among female sex workers in Rajshahi city, Bangladesh: a cross sectional study"

Dear Editor,

Thank you very much for sending your and reviewers’ comments on our manuscript. We have modified and revised the manuscript accordingly, and detailed corrections point–by-point is given below:
Editor Comments:

As it is not a complete survey of FSW in the study area, the title would better to be written as "Knowledge of tuberculosis among female sex workers in Rajshahi city, Bangladesh: a cross sectional study"

Author’s response: Thank you very much for your comments on our manuscript. We have changed the title of this paper according to your comment. The new title is, "Knowledge of tuberculosis among female sex workers in Rajshahi city, Bangladesh: a cross sectional study"

Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Author’s response: Thank you very much for your suggestions on our manuscript. Our manuscript has been revised in English by a local person who is fluent in English.

We would like to thank the editor and reviewers for the valuable comments. We have revised the documents to the best of our ability, but we will definitely be happy to provide further improvement if there are further clarifications required.

With best regards

Dr. Md. Golam Hossain

Professor of Health Research Group

Department of Statistics, University of Rajshahi

Rajshahi-6205, Bangladesh

E-mail: hossain95@yahoo.com